Author's response to reviews

Title: Impact of viral replication inhibition by entecavir on peripheral T lymphocyte subpopulations in chronic hepatitis B patients

Authors:

Jing You (jingyoukm@126.com)
Hutcha Sriplung (hutchas@psu.ac.th)
Alan Geater (alan.q@psu.ac.th)
Virasakdi Chongsuvivatwong (cvirasak@medicine.psu.ac.th)
Lin Zhuang (zlinkm@126.com)
Yun-Li Li (liyunli@yahoo.com)
Hua Lei (hualeikm@yahoo.com)
Jun Liu (liujun@yahoo.com)
Hong-Ying Chen (wangcongyunn@ynao.ac.cn)
Bao-Zhang Tang (t-baozhang@163.com)
Jun-Hua Huang (huangjunhua99@yahoo.com.cn)

Version: 7 Date: 1 July 2008

Author's response to reviews: see over
Dear Editor:

Thank you very much for giving us the opportunity to proof-read and revise and resubmit our manuscript (MS: 1378375013171503, Title: Impact of viral replication inhibition by entecavir on peripheral T lymphocyte subpopulations in chronic hepatitis B patients). We have checked over our manuscript very carefully and found one point that needed correction as below.

Detailed revising point:

Fifth page, “Peripheral blood T-lymphocyte subsets measurement”:
line 6,

error: …of CD3⁺, CD4⁺ and CD8⁺cells found to be...
correction: …of CD3⁺, CD4⁺ and CD8⁺ T-cells found to be...

Thanks again so much.

Yours sincerely,

Jing You
Epidemiology Unit, Faculty of Medicine,
Prince of Songkla University,
Hat Yai, Songkhla 90110, Thailand,
Fax: 66-74-212900
Telephone: 66-74-429754
E-mail: jingyoukm@126.com